

# **JGR** Biogeosciences

# **RESEARCH ARTICLE**

10.1029/2020JG006190

#### **Special Section:**

Winter limnology in a changing world

#### **Key Points:**

- Positive relationship between methane oxidation and phosphorus observed among lakes was supported by a phosphorus amendment experiment
- The methanotrophic community composition was not affected by phosphorus amendments
- Enhanced methane oxidation in nutrient-rich lakes may neutralize more methane trapped under the ice and reduce ice-off emissions

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

**Correspondence to:** 

H. O. Sawakuchi, henrique.sawakuchi@liu.se

#### **Citation**:

Sawakuchi, H. O., Martin, G., Peura, S., Bertilsson, S., Karlsson, J., & Bastviken, D. (2021). Phosphorus regulation of methane oxidation in water from ice-covered lakes. *Journal* of *Geophysical Research: Biogeosciences*, *126*, e2020JG006190. https://doi. org/10.1029/2020JG006190

Received 2 DEC 2020 Accepted 14 AUG 2021

#### **Author Contributions:**

Conceptualization: Stefan Bertilsson, David Bastviken Data curation: Henrique O. Sawakuchi, Gaëtan Martin Formal analysis: Henrique O. Sawakuchi, Gaëtan Martin Funding acquisition: Sari Peura, Stefan Bertilsson, Jan Karlsson, David Bastviken

#### © 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Phosphorus Regulation of Methane Oxidation in Water From Ice-Covered Lakes

# Henrique O. Sawakuchi<sup>1,2</sup>, Gaëtan Martin<sup>3</sup>, Sari Peura<sup>3</sup>, Stefan Bertilsson<sup>4</sup>, Jan Karlsson<sup>2</sup>, and David Bastviken<sup>1</sup>

<sup>1</sup>Department of Thematic Studies–Environmental Change, Linköping University, Linköping, Sweden, <sup>2</sup>Department of Ecology and Environmental Science, Climate Impacts Research Centre (CIRC), Umeå University, Umeå, Sweden, <sup>3</sup>Department of Forest Mycology and Plant Pathology, Science for Life Laboratory, Swedish University of Agricultural Sciences, Uppsala, Sweden, <sup>4</sup>Department of Aquatic Sciences and Assessment, Swedish University of Agricultural Sciences, Uppsala, Sweden

**Abstract** Winter methane ( $CH_4$ ) accumulation in seasonally ice-covered lakes can contribute to large episodic emissions to the atmosphere during spring ice melt. Biological methane oxidation can significantly mitigate such  $CH_4$  emissions, but despite favorable  $CH_4$  and  $O_2$  concentrations,  $CH_4$  oxidation appears constrained in some lakes for unknown reasons. Here we experimentally test the hypothesis that phosphorus (P) availability is limiting  $CH_4$  oxidation, resulting in differences in ice-out emissions among lakes. We observed a positive relationship between potential  $CH_4$  oxidation and P concentration across 12 studied lakes and found an increase in  $CH_4$  oxidation. Hence, while an increase in sedimentary  $CH_4$  production and ebullitive emissions may happen with eutrophication, our study indicates that the increase in P associated with eutrophication may also enhance  $CH_4$  oxidation. The increase in  $CH_4$  oxidation may hence play an important role in nutrient-rich ice-covered lakes where bubbles trapped under the ice may to a greater extent be oxidized, reducing the ice-out emissions of  $CH_4$ . This may be an important factor regulating  $CH_4$  emissions from high latitude lakes.

**Plain Language Summary** Methane produced by microorganisms in lake sediments accumulates under the ice in seasonally ice-covered lakes and is released into the atmosphere when the ice melts. A specialized group of bacteria can consume methane in the water and significantly reduce methane emissions. In some lakes, however, this consumption is limited for unclear reasons. A recent study suggests that phosphorus, an essential nutrient for plants and algae in aquatic environments, could cause this limitation. Here, we tested the influence of phosphorus on the bacterial consumption of methane in 12 lakes with different phosphorus concentrations and made experiments with the addition of phosphorus. Our results indicate that methane consumption was higher in lakes with more phosphorus in the water and that phosphorus addition enhanced the consumption of methane. Despite higher production of methane expected from nutrient-rich lakes, our study shows that increased consumption of methane may reduce part of the enhanced production, limiting the emissions. This neutralization may be higher in seasonally frozen lakes where methane is trapped under the ice, allowing more time for methane consumption.

# 1. Introduction

Lakes are important global sources of  $CH_4$  to the atmosphere. Climate change, especially in northern lakes, may increase  $CH_4$  emissions by up to 54% (Wik et al., 2016). Furthermore, this increase may be exacerbated by eutrophication (Beaulieu et al., 2019; DelSontro et al., 2018). Nearly half of the global lake surface area is located in temperate and boreal zones where lakes are ice-covered for some period in the winter (Downing et al., 2006; Matthews et al., 2020). Under the ice,  $CH_4$  bubbles released from the sediment can accumulate throughout the winter and result in high ice-out release in spring (Wik et al., 2016).

The observations of  $CH_4$ -rich bubbles under lake ice at the end of the winter and associated emissions upon ice-out are mysterious for several reasons. (a) This  $CH_4$  is exposed to oxic water for extended periods, which should create favorable conditions for  $CH_4$  oxidation, known to have the capacity to deplete water column



**Investigation:** Henrique O. Sawakuchi, Gaëtan Martin

**Methodology:** Henrique O. Sawakuchi, Gaëtan Martin, Sari Peura, Stefan Bertilsson, Jan Karlsson, David Bastviken

Project Administration: Henrique O. Sawakuchi Visualization: Henrique O.

Sawakuchi, Gaëtan Martin

 $\rm CH_4$  to very low levels near the oxycline and found to be more sensitive to substrate availability than temperature (Bastviken et al., 2008; Duc et al., 2010; Kankaala et al., 2006). (b) Bubbles trapped under the ice spend a long time in the water, and over time a small fraction (8%) can be encapsulated and stored in growing ice or exchange gases with the water, dissolving towards equilibrium (80%), where it can be oxidized (Greene et al., 2014; Sepulveda-Jauregui et al., 2015). (c) Ebullition from sediments should decrease during the winter as shallow sediments cool down and if the availability of labile organic substrates from primary production fueling methanogenesis get progressively reduced (Wik et al., 2018). This production of  $\rm CH_4$  in sediment is the main  $\rm CH_4$  source in most lakes, except for the limited number of lakes experiencing continuous seepage of  $\rm CH_4$  of geogenic origin or from deep melting permafrost (Walter et al., 2008). Therefore, extensive renewal of under-ice  $\rm CH_4$  bubbles in the late phase of the ice-covered winter period is unlikely in most lakes. Despite this,  $\rm CH_4$ -rich water under the ice and high ice-out emissions have been observed in some lakes (Sepulveda-Jauregui et al., 2015).

Recently, a study found an unexpected absence of  $CH_4$  oxidation in water from ice-covered lakes, while  $CH_4$  oxidation was extensive in other nearby lakes, highlighting that information regarding the factors controlling winter  $CH_4$  oxidation in lakes is still scarce and contradictory (Denfeld et al., 2018). Martinez-Cruz et al. (2015) observed that the winter  $CH_4$  oxidation variability was mainly related to  $O_2$  availability in Alaskan lakes, which is in agreement with previous observations (Bastviken, 2009; Segers, 1998), but cannot explain the range of potential  $CH_4$  oxidation found in oxic surface waters under the ice (Denfeld et al., 2016). It has been suggested that high  $O_2$  concentration may inhibit  $CH_4$  oxidation (Reis et al., 2020), and light is another factor suggested to inhibit  $CH_4$  oxidation (Dumestre et al., 1999; Murase & Sugimoto, 2005). However, a persistent ice cover may both reduce the  $O_2$  and attenuate light below inhibitory levels in the water column. Under such conditions, Denfeld et al. (2016) found that the lakes where  $CH_4$  oxidation was absent. Phosphorus (P) is a major nutrient required for all life forms (Westheimer, 1987) and could potentially limit methanotrophic activity in lakes (Denfeld et al., 2016). Further, correlation between  $CH_4$  oxidation and P concentration has been previously observed in soils (Chauhan et al., 2012; Gray et al., 2014).

Clearly, under-ice  $CH_4$  oxidation can be an important constraint for ice-off  $CH_4$  emissions, but for unknown reasons,  $CH_4$  oxidation activity seems to vary widely among lakes. The present study aimed to estimate  $CH_4$  oxidation in lakes covering a range of total P availability and experimentally assess cause-effect relationships between P concentration and  $CH_4$  oxidation to test the hypothesis that P availability favors  $CH_4$  oxidation at low-temperature conditions representative of under-ice conditions in boreal lakes.

# 2. Materials and Methods

#### 2.1. Lakes and Sampling

We visited 12 lakes in southeast Sweden in March 2018 and collected 4.5 liters of water 10 cm below the ice after carefully drilling the ice with a manual ice auger. Water was sampled at the deepest point for lakes with bathymetric information and near the center when depth information was not available. Ice thickness ranged from 24 to 41 cm. Lakes size ranged from 0.02 to 23.7 km<sup>2</sup> and have different trophic states (oligotrophic, mesotrophic, and eutrophic). The ice-cover period for these lakes is between December and April (SMHI, 2006). Seven of the lakes had been included in a previous study addressing the impact of winter conditions on  $CH_4$  oxidation (Denfeld et al., 2016).

Samples (5 ml) for the in-situ concentration of  $CH_4$  were collected with a syringe connected to a tube placed under the ice and immediately transferred to 20 ml glass vials filled with high purity nitrogen gas  $(N_2)$  and with 150  $\mu$ L  $H_3PO_4$  (85%) to preserve the sample. Water samples were also collected for total P (TP) and dissolved organic carbon (DOC). DOC samples were filtered through combusted GFF filters into glass vials. Non-filtered water samples were collected in acid-washed (HCl 10%) HDEP bottles and preserved with  $H_2SO_4$  for TP analysis. Water for the incubation experiments was pumped directly from the lake into acid-washed cubitainers (4.5 L) using a battery-operated Masterflex peristaltic pump (Cole-Parmer) and kept at 4°C. Water temperature, pH, dissolved oxygen, and electric conductivity were measured using multiparameter probes (YSI EXO2 and Aquaread AP-5000). Lake water was analyzed for DOC using a



10.1029/2020JG006190

TOC analyzer (Shimadzu TOC-V<sub>CSH</sub>). TP and PO<sub>4</sub> <sup>3–</sup> were analyzed colorimetrically using an AutoAnalyzer (BRAN + LUBBE AutoAnalyzer3).

#### 2.2. Methane Oxidation Assessment and P Amendment

Acid washed (10% HCl) and combusted (550°C for 4h) 120 ml glass vials were rinsed and filled with 80 ml of lake water. For each lake, 10 vials were prepared using water from the same cubitainer, and 250 µL of an aqueous  $PO_4^{3-}$  solution (NaH<sub>2</sub>PO<sub>4</sub>\*H<sub>2</sub>O) was added in five of them to increase the  $PO_4^{3-}$  concentration by 500 µg/L. In total, 120 vials were incubated for the experiment. Vials were capped with acid-washed and autoclaved 10 mm butyl rubber stoppers and sealed with aluminum crimps. Before incubation, vials' headspaces were flushed for 5 min with high purity carbon-free synthetic air to remove CH<sub>4</sub> and CO<sub>2</sub> and standardize the concentration of O<sub>2</sub>. After that, 1.8 ml of 99% CH<sub>4</sub> was added to the headspace to reach a  $CH_4$  concentration of 100  $\mu$ M  $CH_4$  in the water at 4°C, according to Henry's law. Incubations were carried out in the dark at 4°C on a shaking table to gently mix the water 15 min every two hours. Previous similar experiments to assess the potential CH<sub>4</sub> oxidation in water from ice-covered lakes incubated at 2-4°C have reported a delayed start (lag-phase) varying from 7 to 12 days (Canelhas et al., 2016; Martinez-Cruz et al., 2015). Thottathil et al. (2019) observed that lake water incubations to measure  $CH_4$  oxidation at low temperatures (5°C) require incubations over several weeks before a significant change in  $CH_4$  concentration can be detected. Here, CH<sub>4</sub> headspace concentration in the vials were monitored during 83–88 days and samples taken approximately every two weeks. Long-term incubations (up to 90 days) were previously done to evaluate the influence of light and inorganic nitrogen on CH<sub>4</sub> oxidation (Murase & Sugimoto, 2005).

Gas samples from the headspace in the vials were taken by adding, mixing, and withdrawing 3 ml of synthetic air, using a 5 ml syringe, for analysis by gas chromatography (GC). This procedure was done to avoid pressure changes during the experiment and allow monitoring the CH<sub>4</sub> concentration over time in the same vial. Dilution effects by sampling were corrected based on blank control vials prepared with Milli-Q water and run along with the lake water vials throughout the whole experiment. In total, 12 blank vials were prepared with the same amount of CH<sub>4</sub>, and with or without the addition of PO<sub>4</sub><sup>3-</sup>, and monitored together. Thus, the decrease in CH<sub>4</sub> in the blank vials was due to dilution caused by the addition of synthetic air when sampling. Analysis of CH<sub>4</sub> for all samples was made using a GC with a flame ionization detector and a methanizer (Agilent 7890; 60/100 poropakQ, 6 ft × 1/8 in column, manual injection via a sample loop), with a detection limit of  $4.4 \times 10^{-7}$  mmol of CH<sub>4</sub>.

To calculate  $CH_4$  oxidation based on the decline in headspace  $CH_4$  over time, we used the decay constant ( $\lambda$ ) as the specific rate of methane oxidation (SRMOx) according to the following equation:

$$C_t = C_0 e^{-\lambda t} \tag{1}$$

 $\lambda$  is the decay constant representing the fraction of the CH<sub>4</sub> being oxidized per time unit (day<sup>-1</sup>; dimensionless),  $C_t$  is the concentration at time *t* (day), and  $C_0$  is the starting concentration after the lag-phase.  $\lambda$  was determined from the best fit of Equation 1 relative to the time series of measurements in each incubation bottle. The fit error was estimated using the R<sup>2</sup> of the linear fit after log transformation.

For each vial, we considered that a lag-phase occurred during the analysis period when the concentration in the vial remained inside the confidence interval (95%) of the starting  $CH_4$  concentration. A lag-phase of 12–23 days was seen in most vials. However, cases without lag-phase or where no  $CH_4$  oxidation was recorded were also observed. We focused on the period of decline even if occurring at different times in different bottles. Hence, the  $CH_4$  oxidation capacity of each vial, was calculated using the time after the lag-phase period.

Our incubations had a large reservoir of  $O_2$  in the air headspace volume (40 ml headspace and 80 ml water). The aeration of the vials by air purging at the start of the experiment equilibrated the water with  $O_2$  in the air, leading to an expected  $O_2$  concentration of approximately 13 mg/L in all vials. Assuming that the decrease in  $O_2$  is mainly attributed to DOC degradation we estimated the maximum expected  $O_2$  decrease during incubation as follows: A previous long-term incubation (426 days) of lake water representing a similar range of DOC and TP levels as in this study (including one of the lakes in this study, Lillsjön) at 15°C, found a total DOC decrease of 30% (Bastviken et al., 2004). If assuming a 1:1 molar  $O_2$  consumption per DOC-C mineralized, a 30% DOC mineralization with our highest starting DOC concentration (33.1 mg/l)



would have depleted a maximum of 10% of the  $O_2$  reservoir in our vials. Our incubation over much shorter time (88 days) and at 4°C makes such high mineralization and  $O_2$  consumption unrealistic and a more likely maximum  $O_2$  consumption is in the order of 2.5% during the incubation (assuming a factor of 2 reduction from the time difference and another factor of 2 reduction from the temperature difference). Hence, the  $O_2$  reservoir in the vial headspace kept the  $O_2$  levels high and similar across all vials.

#### 2.3. Microbial Community Characterization

To assess the impact of the P amendment on the microbial communities and test whether differences in SRMOx may be explained by change in the methanotroph communities we sampled and analyzed independently each replicate at the end of the incubation. For that purpose, 40 ml of water were sampled from each vial and stored at  $-20^{\circ}$ C for further DNA analysis. Once thawed, each sample was filtered through separate 0.22 µm Durapore<sup>®</sup> Membrane Filters (Merck, Darmstadt, Germany). Filters were stored overnight at  $-20^{\circ}$ C and DNA extracted using the DNeasy PowerSoil Kit (Quiagen, Hilden, Germany). Library preparation for 16S rRNA gene analysis was done following a two-steps Polymerase Chain Reaction (PCR) protocol, as described in Sinclair et al., 2015. All PCRs were conducted in 20 µl of volume using 0.02 U/µl Phusion high fidelity DNA polymerase, 1X Q5 reaction buffer (NEB, UK), 0.25 µM primers and 200 µM dNTP mix and 1 µl DNA template.

The first step was performed in triplicate with primers 341F (3'-CACTCTTTCCCTACACGACGCTCTTC-CGATCTNNNNCCTACGGGNGGCWGCAG-5') and 805NR (3'-AGACGTGTGCTCTTCCGATCTGACTAC-NVGGGTATCTAATCC-5') (Herlemann et al., 2011). The thermal program consisted of 20 cycles with an initial 98°C denaturation step for 10 min, a cycling program of 98°C for 10 s, 48°C for 30 s, and 72°C for 30 s and a final elongation step at 72°C for 2 min. Triplicate PCR reactions were then pooled and purified with magnetic beads (Sera-Mag<sup>™</sup> Select, GE Healthcare, Chicago, United States of America), and 2 µl of the purified products were used at a template for a second stage PCR, where indexed primers were added. The second thermal program consisted of 15 cycles with an initial 98°C denaturation step for 30 s, a cycling program of 98°C for 10 s, 66°C for 30 s, and 72°C for 30 s and a final elongation, PCR products were again purified with magnetic beads and quantified with Qubit<sup>™</sup> using the Qubit<sup>™</sup> dsDNA HS Assay Kit (Invitrogen<sup>™</sup>). Finally, 15.6 µg of each indexed and purified PCR product were pooled before submission of the sample to the Science for Life Laboratory SNP/SEQ sequencing facility hosted by Uppsala University (Uppsala, Sweden). Sequencing was done using Illumina Miseq in paired-end mode with 300bp and v3 chemistry.

Sequence processing was performed with Mothur 1.41.0 following the MiSeq SOP (Kozich et al., 2013), with the exception that clustering to operational taxonomic units (OTUs) was done using VSEARCH (Rognes et al., 2016) as implemented in Mothur. The OTU table, consensus taxonomy table, and a phylogenetic tree created using Mothur were analyzed in R using the phyloseq package (McMurdie & Holmes, 2013). Rarefaction was applied to the OTU table before further analysis. Sampling size for the rarefaction was set at 90% of the number of sequences present in the sample with the lowest amount of sequences, and set. seed (1) was used to initialize repeatable random subsampling, retained for further analysis. Following rarefaction, all OTUs with less than 10 reads in the subsampled data were removed from the OTU table. OTUs were considered methanotrophic if they were assigned to one of the known aerobic methanotrophic taxa. Those taxa are the orders Methyloccella, Methyloferula, Candidatus Methylomirabilis, and Candidatus Methanoperedenaceae.

#### 2.4. Data Analysis

The overall effect of lake TP in SRMOx was tested by linear regression analysis using log-transformed SR-MOx to achieve normality and homogeneity of variance of residuals. The difference between non-amended and P-amended incubations in each lake was assessed by a two-sided Wilcoxon rank-sum test. The potential influence of in situ water chemistry parameters on the SRMOx was assessed by the two-sided Spearman correlation test using SRMOx observed in non-amended incubations. Linear regression, Wilcoxon rank-sum test, and Spearman correlation test were performed in R using the built-in Stats Package (R Core



#### Table 1

In-Situ Concentrations of Dissolved Molecular Oxygen (DO), Dissolved Organic Carbon (DOC), Methane (CH<sub>4</sub>), Total Phosphorus (TP), and in the Water 10 cm Below the Ice

Lake ID	Lakes	DO (mg/L)	DOC (mg/L)	СН <sub>4</sub> (µМ)	TP (µg/L)				
MOX01 <sup>a</sup>	Lumpen	6.7	25.2	0.08	14.9				
MOX02 <sup>a</sup>	Björklinge-Långsjön	11.8	2.6	1.14	9.9				
MOX03 <sup>a</sup>	Erken	13.4	4.9	0.12	24.6				
MOX04 <sup>a</sup>	Malstasjön	3.8	9.9	0.07	74.3				
MOX05 <sup>a</sup>	Fyrsjön	8.6	6.3	0.07	10.5				
MOX06 <sup>a</sup>	Plåten	3.0	23.6	0.04	14.7				
MOX07 <sup>a</sup>	Lötsjön	8.0	5.2	0.06	24.9				
MOX08	Glimmingen	11.3	8.3	0.11	5.3				
MOX09	Stortoveln	10.6	7.1	0.24	5.3				
MOX10	Bengtsgolen	0.0	33.1	0.14	34.7				
MOX11	Lillsjön	5.9	25.5	0.16	15.9				
MOX12	Parsen	4.2	16.7	0.1	15.8				
<sup>a</sup> Lakes where under-ice CH <sub>4</sub> oxidation was evaluated in Denfeld									

et al., <mark>2016</mark>.

Team, 2019). To test the effect of the lake of origin and treatment on total and methanotrophic communities, we used PERMANOVA (Adonis, vegan package, R) based on Bray distance matrices, with 10,000 free permutations and number of groups minus one degree of freedom. When the p-value was between 0.1 and 0.001, we ran 10<sup>6</sup> extra permutations.

## 3. Results

In-situ CH<sub>4</sub> concentrations were higher than atmospheric saturation (~5 nM at 0°C) for all lakes, with concentrations ranging from 0.04 to 1.1 µM (Table 1). Total P (TP) concentrations in the studied lakes ranged from 5.3 to 74.3  $\mu$ g/L, classifying the lakes as oligotrophic, mesotrophic, and eutrophic (Tables 1 and S1). Dissolved O<sub>2</sub> concentration also varied between lakes, but only one lake (MOX10, Bengtsgolen) was anoxic. There were large variability in DOC (2.6-33.1 mg/L, Table 1) and other in-situ physicochemical parameters (Table S1) among the lakes. During the incubation, with exception of Lake MOX03, the change in CH<sub>4</sub> concentrations in the P-amended vials was larger than in the non-amended. The largest change in CH<sub>4</sub> concentration was observed in MOX04 vials where 98.7% and 99.9% of the initial amount of CH, were consumed in the non-amended and P-amended vials, respectively, with the lowest concentration reaching  $2 \times 10^{-5}$  mmol of CH<sub>4</sub> (Table S2). For individual lakes, we observed a large variability of SRMOx ranging from 0 (no CH<sub>4</sub> oxidation registered for Lake MOX11) to 0.09 days<sup>-1</sup> for non-amended

vials and from 0.006 to 0.306 days<sup>-1</sup> for P-amended vials. The overall average fit error based on the R<sup>2</sup> of the linear fit of log transformed data was 0.93 ± 0.09, ranging from 0.50–1.00 for individual vials. Despite the consistently higher mean SRMOx in P-amended vials for most of the lakes, the difference was significant for only four lakes (Wilcoxon, p < 0.05) (Table S3). Methane oxidation rate ranged from 2.5–13.5 mmol m<sup>-3</sup> d<sup>-1</sup> and 3.9–29.4 mmol m<sup>-3</sup> d<sup>-1</sup> for non-amended and P-amended, respectively (Table S4).

Overall, higher SRMOx was observed in P-amended incubations compared to the non-amended ones  $(0.041 \pm 0.094 \text{ days}^{-1} \text{ and } 0.014 \pm 0.027 \text{ days}^{-1}$ , respectively; Wilcoxon, p < 0.001). Single lake evaluation shows an enhancement in P-amended vials in 11 out of the 12 studied lakes, the only exception was for MOX03, where higher SRMOx was observed in the non-amended vials (Figure 1, Table S3). However, despite the overall enhancement in CH<sub>4</sub> oxidation attributed to P amendment in the single lakes, in approximately half of the lakes, not all P-amended replicates showed an increase in SRMOx, resulting in five lakes where the difference was statistically significant.

In addition to the P-amendment experiment we observed a positive relationship between the natural (i.e., non-amended) SRMOx and in-situ TP concentration ( $R^2 = 0.71$ , p = 0.001, Figure 2). Despite that the highest TP and SRMOx observed in lake MOX04 seems to be driving the observed trend, a significant positive effect remains if this lake is removed from the evaluation ( $R^2 = 0.50$ , p = 0.022). However, we acknowledge that more data from eutrophic and hypereutrophic lakes would have improved our evaluation. We did not find any significant effect between DOC and SRMOx ( $R^2 = 0.006$ , p = 0.82).

After trimming and clustering the number of reads in each sample ranged from 43,452 to 100,884 (Table S5). After rarefaction and removal of rare OTUs (represented by less than 10 reads across all samples), the combined data set consisted of 4,692,720 reads grouping into 2,663 OTUs, used for further analyses. Methanotrophs represented 0.02%–69.4% of all reads in individual samples (Figure S2). Methanotroph reads were assigned to 228 OTUs almost exclusively affiliated with Gammaproteobacteria (99.3% of the analyzed methanotroph reads), where 94.5% were accounted by the five most abundant OTUs. The five most abundant OTUs were closely related to the two cultivated psychrophilic methanotrophs belonging to the Methylococcaceae family, *Methylobacter tundripaludum* and *Methylovulum psychrotolerans*, with 16S rRNA sequence identities ranging from 95.5% to 99.6% (Table 2). The proportion of reads assigned to methanotrophs in relation to the total microbial community reads is what we call proportion of methanotrophs.





**Figure 1.** Comparison of the specific methane oxidation rate (SRMOx) observed among non-amended (N) and P-amended (P) vials.  $CH_4$  oxidation was not observed in non-amended vials of MOX11. Note that the *y*-axis scale is adjusted for the values observed for each lake. Dots show the SRMOx values of replicates where  $CH_4$  oxidation was observed. The thick line in the middle of the box is the median, and the lower and upper boundaries represent 25th and 75th percentiles, points outside the box, and the error bar correspond to  $1.5^*$  inter-quartile range from the hinge. Symbols on the right upper corner of the panels represent the p-value ranges (0–0.001 = '\*\*\*'; 0.01–0.05 = '\*'; 0.05–0.1 = '+'; 0.1–1.0 = "No symbol"), based on Two-sided Wilcoxon rank-sum test presented on the Table S3.



**Figure 2.** Linear relationship between the log-transformed mean specific rates of methane oxidation (SRMOx) for non-amended vials and in-situ total phosphorus (TP) concentrations. The error bar shows the standard deviation of the mean for the replicates.



#### Table 2

Similarity of the Most Common Operational Taxonomic Units With Their Closest Isolated Organism and Their Occurrence Across Vials, Treatments, and Lakes

				Occurrence		
OTU	Closest cultivated relative	Identity (%)	Proportion of reads (%)	All vials	Treatment	Lake
otu_0003	Methylobacter tundripaludum	98.7	36.0	118/120	24/24	12/12
otu_0004	Methylovulum psychrotolerans	96.8	31.3	99/120	24/24	12/12
otu_0010	Methylovulum psychrotolerans	99.4	13.3	96/120	23/24	10/12
otu_0014	Methylobacter tundripaludum	95.5	7.7	86/120	24/24	12/12
otu_0018	Methylobacter tundripaludum	95.5	6.6	71/120	21/24	10/12

*Note.* Identity indicates the similarity of the OTU with a cultivated organism based on 16S rRNA gene amplicons. The proportion of reads is relative to the total number of reads (789,759) identified as methanotrophs.

The overall relative proportion of methanotrophs in the non-amended incubations was not correlated to the SRMOx or the lake TP concentration. We also did not find any difference in the average proportion of methanotrophs between P-amended and non-amended vials ( $0.20 \pm 0.18$  and  $0.14 \pm 0.16$ , respectively; Wilcoxon, p = 0.087).

Analysis of the integrated influence of lake and treatment illustrated by non-metric multidimensional scaling (NMDS) indicate an influence on the total bacterial community composition (PERMANOVA p < 0.001) with a stronger influence of the lake of origin on the bacterial community composition than for the P-amendment, as reflected by their respective  $R^2$  values of 0.42 and 0.03 (Figure 3a). When only the



Figure 3. Overall beta-diversity for total bacterial and methanotrophic community. Beta-diversity analysis using non-metric multidimensional scaling (NMDS) for total bacterial (a) and methanotrophic (b) communities.

methanotrophs are considered, only the lake ID has an effect on the community composition (p < 0.001,  $R^2 = 0.34$ , F = 5.34), whereas treatment has no effect (p = 0.37,  $R^2 = 0.01$ , F = 1.07) (Figure 3b).

Single lake evaluation of the total bacterial beta-diversity suggested that for seven lakes, the addition of P affected the community composition as the samples tended to group according to treatment with ADONIS *p*-value below 0.05 (Figure S1a).

Even if NMDS plots suggested an effect of P-amendment on the methanotrophic communities in lakes MOX10 and MOX11, all ADONIS *p*-values were above 0.05 (Figure S1b).

#### 4. Discussion

The observed  $CH_4$  and  $O_2$  levels correspond to previously reported values from ice-covered lakes (Denfeld et al., 2016; Samad & Bertilsson, 2017; Sepulveda-Jauregui et al., 2015). SRMOx of 0.074 days<sup>-1</sup> reported for under-ice water samples from Lake Erken (Canelhas et al., 2016) was within the overall range of SR-MOx observed in our study, but this value is higher than our SRMOx estimate for the same lake (MOX03). Previously reported summertime surface water SRMOx for temperate and boreal lakes varying from 0.04 to 0.94 days<sup>-1</sup> was, in general, higher than what we observed in winter (Bastviken et al., 2008; Thottathil et al., 2019). However, seasonal variation indicating higher  $CH_4$  oxidation in summer compared with winter has already been reported for arctic lakes (Martinez-Cruz et al., 2015), yet our potential methane oxidation rates were within the range (0.01–36.1 mmol m<sup>-3</sup> d<sup>-1</sup>) previously observed in incubations at low temperature (i.e., 2°C–5°C) for northern lakes (Martinez-Cruz et al., 2015; Sepulveda-Jauregui et al., 2018; Thottathil et al., 2018).

We found a similar variability in  $CH_4$  oxidation among the lakes previously studied by Denfeld et al. (2016), with increasing  $CH_4$  oxidation from MOX07 (Lötsjön), MOX03 (Erken), and MOX04 (Malstasjön). Variability in  $CH_4$  oxidation among 30 ice-covered lakes was also observed in Alaska, where the causes of such variability were attributed to  $CH_4$  or  $O_2$  limitations (Martinez-Cruz et al., 2015). Thottathil et al. (2018) found a positive correlation between  $CH_4$  oxidation and DOC for 14 boreal lakes during the summer, however, this correlation was argued to be attributed to indirect factors such as light inhibition and  $O_2$  availability, which agrees with the lack of correlation between DOC and SRMOx in this study. In our experiment, both  $CH_4$ and  $O_2$  were amended to the vials at the start of the incubation that was carried out in the dark. Hence, the variability observed among our lakes was not expected to be related to lack of substrate or light inhibition.

The positive relationship we observed between SRMOx and in-situ TP concentrations in the lakes supports the potential influence of P on  $CH_4$  oxidation pointed out by Denfeld et al. (2016). Previously, correlation between  $CH_4$  oxidation and P has also been reported in Everglade dry wetlands soils (Chauhan et al., 2012), arctic permafrost soils (Gray et al., 2014), tropical forest soils (Zhang et al., 2011), and in temperate drainage ditches (Veraart et al., 2015). In addition to the positive relationship between SRMOx and P in natural conditions, our experimental results suggest an enhancement of SRMOx attributed to P amendment, providing mechanistic support to the positive relationship between SRMOx and lake TP concentration below the ice (Figures 1 and 2).

Despite a change in the whole microbial community composition observed between treatments, the methanotrophic community composition seemed unaffected by P, regardless of lake, and was systematically dominated by only a few OTUs closely related to psychrophilic methanotrophs (Table 2). The dominance of Methylococcaceae in the methanotrophic community at the end of our incubation experiment is in line with previous assessments of winter  $CH_4$  oxidation in some of our studied lakes, where a significant increase in relative abundance of Methylococcaceae was observed during incubations (Canelhas et al., 2016; Denfeld et al., 2016). The dominance of Type I methanotrophs in previous studies where a phosphorus- $CH_4$ oxidation relationship was observed (Gray et al., 2014; Veraart et al., 2015) is also in line with our findings. The dominance of these type I psychrophilic methanotrophs has also been observed in other boreal and arctic freshwater bodies (Crevecoeur et al., 2019; Rissanen et al., 2018) and from other different cold environments like groundwater, northern taiga, tundra soils, polar lakes and permafrost sediments (Kalyuzhnaya et al., 1999; Khmelenina et al., 2002; Trotsenko & Khmelenina, 2005; Vecherskaya et al., 1993).



Regardless the positive correlation observed between CH<sub>4</sub> oxidation and TP concentration we did not find a significant correlation between the relative abundance of methanotrophs and SRMOx or TP concentration. This, combined with the dominance of an OTU associated with psychrophilic methanotrophs, suggests that low temperature and possibly high CH<sub>4</sub> and O<sub>2</sub> concentrations might have a stronger effect than P on the methanotrophic community selection. The observed lack of correlation between the relative abundance of methanotrophs and SRMOx could be attributed to a similar increase in the abundance of all bacteria, resulting in unchanged relative abundance. The observed increase in CH<sub>4</sub> oxidation attributed to P availability could be related to an increase in the absolute abundance, despite the absence of relative increase of methanotrophs in P amended samples, yet we do not have the data to verify this. It is also possible that only the activity of methanotrophs was increased. Several studies where nutrient concentrations have been manipulated report a decoupling of bacterial production and abundance (DeBruyn et al., 2004; Vrede et al., 1999), suggesting that a system could respond with an increase in production, or other processes like CH<sub>4</sub> oxidation, without any parallel increase in their abundances. This decoupling was observed for MOB in a boreal lake and in a rice paddy where CH<sub>4</sub> oxidation did not seem to be attributed to MOB abundance (van Grinsven et al., 2021; Zheng et al., 2013), suggesting that the simple change in activity could also explain the increased CH<sub>4</sub> oxidation observed in our experiment.

The effect of P on the SRMOx could be indirect via stimulation of the combined microbial community, rather than a direct and specific P limitation experienced by MOB. It was indeed pointed out that methanotrophs may benefit from the release of compounds, such as vitamins, amino acids, and organic acids by other bacteria (Iguchi et al., 2011; Xing et al., 2006), while non-methanotrophic bacteria could benefit from enhanced exopolysaccharide production by type I methanotrophs under high nutrient (e.g., P) and substrate availability (Malashenko et al., 2001). This potential systemic effect of the combined microbial community and mutualistic interactions was also observed in experiments showing that MOB (*Methylomonas spp.*) increased their activity (consuming more  $CH_4$ ) when the richness of heterotrophic bacteria increased (Ho et al., 2014). The increase in activity observed in our experiment could be then attributed to a systemic effect of P on the microbial community methanotroph activity rather that a direct use of P by methanotrophs. However, if neither indirect or direct P effects can be excluded, our results suggest that the influence of such interactions and secondary metabolites on  $CH_4$  oxidation could be important in nutrient-rich lakes where a more diverse microbial community could be sustained, in relation to lakes with lower trophic state (Kiersztyn et al., 2019).

We acknowledge that the lack of absolute abundance of methanotrophs limited a clear understanding about to what extent the increase in CH<sub>4</sub> oxidation due to P was attributed to an increase in the methanotrophic density or due to an increase in activity. Similarly, a time series would have been beneficial, but the incubation volume and microbial biomass concentration did not allow for multiple DNA sampling timepoint. Additionally, our incubations were set to be similar to the condition observed in the surface water below the ice where O<sub>2</sub> was present in most lakes. Inhibition of CH<sub>4</sub> oxidation by high O<sub>2</sub> concentration have been recently suggested as a factor influencing CH<sub>4</sub> oxidation in lakes (Reis et al., 2020; Thottathil et al., 2019). The O<sub>2</sub> reservoir in the vial headspace kept the O<sub>2</sub> levels high (i.e., most likely never below 97.5% of the starting levels) and similar across all vials. Thereby possible differences in  $O_2$  inhibition of CH<sub>4</sub> oxidation among vials could not have been large enough to explain the results. As O<sub>2</sub> levels were high during incubation we cannot exclude O<sub>2</sub> inhibition of CH<sub>4</sub> oxidation. However, maximum CH<sub>4</sub> oxidation observed at high O<sub>2</sub> saturation suggest that O<sub>2</sub> inhibition is not always happening, and that this potential inhibition remain unclear (van Grinsven et al., 2020). Nevertheless, any potential O2 inhibition did not prevent the P-induced stimulation of the CH<sub>4</sub> oxidation. We also do not discard the possibility that the higher O<sub>2</sub> concentration in our experiment in comparison with in-situ conditions could have stressed the methanotrophic community adapted to low  $O_2$  concentration, limiting their capacity to oxidize  $CH_4$  in the experiment. Further work on how the magnitude of the P-stimulation of  $CH_4$  oxidation is interacting with levels of O<sub>2</sub> or other factors influencing CH<sub>4</sub> oxidation, would be interesting.

Our P amendment experiment indicate that the correlation between SRMOx and TP observed in field data may be causal. Our study further suggests that the increase in potential  $CH_4$  oxidation along the trophic gradient of lakes observed by Sepulveda-Jauregui et al. (2018) could be attributed to the P availability in the water. However, the synergistic effect of increased nutrient concentration and water temperature were found



to intensify the potential  $CH_4$  production in microcosm lake sediment experiments (Sepulveda-Jauregui et al., 2018), and ebullitive  $CH_4$  emissions from mesocosm experiments (Davidson et al., 2018) suggest that despite the higher offset by oxidation,  $CH_4$  emissions could still increase with warming and eutrophication. It is worth highlighting that enhanced  $CH_4$  oxidation can significantly reduce the amount of  $CH_4$  in the water column, reducing the emissions through diffusion. Hence, it is important to note that assessments of  $CH_4$  fluxes from eutrophic systems should be designed to appropriately cover ebullitive emissions of  $CH_4$  to prevent underestimation of the total  $CH_4$  emissions. The P- $CH_4$  oxidation link could be an important factor controlling diffusive fluxes during the open water season. However, we argue that this importance could be more significant during winter when  $CH_4$  oxidation also mitigates ebullition by oxidation of  $CH_4$  in bubbles trapped under the ice and diffusing back into the water.

Similarly, photosynthetic consumption of P could cause P deficiencies in the surface lake water column, leading to decreasing methane oxidation rates. Such a P deficiency, rather than direct light inhibition, could be a complementary or alternative explanation to the proposed enzymatic inhibition of  $CH_4$  oxidation by light (Dumestre et al., 1999; Murase & Sugimoto, 2005; Shelley et al., 2017). An increase in phytoplankton associated with a decrease in the relative proportion of MOB and a consequent increase of in situ  $CH_4$  concentration was observed after the experimental removal of the snow cover on a seasonally ice-covered lake (Garcia et al., 2019). Similarly, MOB expansion in the water column has also been observed during the ice season as compared to summer when MOB were largely restricted to the deeper zones of the lakes (Samad & Bertilsson, 2017). During summer, competition for P would be higher near the surface while P concentrations usually increase in the anoxic zone of the water column.

Wintertime has been highlighted as more important for the carbon cycle than previously assumed (Denfeld et al., 2018; Sharma et al., 2020), and our results bring evidence that P availability could increase winter  $CH_4$  oxidation with high potential for mitigating  $CH_4$  emissions to the atmosphere. The identification of P as a controlling factor of  $CH_4$  oxidation can improve predictions of ice-off and open water emissions and, thus, annual  $CH_4$  emissions from lakes. This study also provides perspectives for the future. The predicted reduction in the length of the ice-covered season and snow coverage (Sharma et al., 2019) could reduce bubble trapping by ice and thereby reduce the mitigative capacity of methane emissions in winter, especially in nutrient-rich (i.e., eutrophic) lakes where ebullition is suggested to be highest (Beaulieu et al., 2019) and P would not limit  $CH_4$  oxidation. This can result in overall higher annual emissions from lakes, and contribute as a additional  $CH_4$  feedback to global warming (Dean et al., 2018).

# 5. Conclusion

Based on the overall positive response of  $CH_4$  oxidation to phosphate addition observed in the experiment, we conclude that when  $CH_4$  and  $O_2$  are not limiting factors,  $CH_4$  oxidation in lakes could be constrained by P availability. We found that limited  $CH_4$  oxidation could be expected during winter in nutrient-poor lakes, potentially explaining why large emissions during ice-off may happen in some lakes but not in others. On the other hand, high P availability could increase  $CH_4$  oxidation capacity, especially in winter, when ebullitive emissions are trapped under the ice and subjected to  $CH_4$  oxidation, with the potential to reduce the ice-out share of the annual emissions. We do not exclude the possibility that other variables alone or in combination with P may also play a role in controlling  $CH_4$  oxidation. We also acknowledge that P may favor  $CH_4$  emissions by increasing eutrophication, which negatively influences  $O_2$  availability due to increased heterotrophic respiration, creating conditions that could limit under ice  $CH_4$  oxidation resulting in higher ice-off emissions. However, despite the seemingly important role P has on winter  $CH_4$  oxidation, its influences on the balance between  $CH_4$  production and  $CH_4$  oxidation needs further investigation across different seasons and lake types, helping understand the regulations of the net  $CH_4$  balance and improve estimates of annual  $CH_4$  emissions from lakes.

# **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.



## Data Availability Statement

Sequences and associated metadata are deposited in NCBI's Sequence Read Archive and BioSample under accession number PRJNA638356 and can be accessed through the link: https://www.ncbi.nlm.nih.gov/bio-project/PRJNA638356. All data used in the paper appear as metadata of each sample.

## References

Bastviken, D. (2009). Methane. In E. L. Editor-in-Chief, & Gene (Eds.), *Encyclopedia of inland waters* (pp. 783–805). Academic Press. https://doi.org/10.1016/b978-012370626-3.00117-4

Bastviken, D., Cole, J. J., Pace, M. L., & Van de Bogert, M. C. (2008). Fates of methane from different lake habitats: Connecting whole-lake

budgets and CH<sub>4</sub> emissions. *Journal of Geophysical Research-Biogeosciences*, *113*(G02024), 1–13. https://doi.org/10.1029/2007jg000608 Bastviken, D., Persson, L., Odham, G., & Tranvik, L. (2004). Degradation of dissolved organic matter in oxic and anoxic lake water. *Limnology & Oceanography*, *49*(1), 109–116. https://doi.org/10.4319/lo.2004.49.1.0109

Beaulieu, J. J., DelSontro, T., & Downing, J. A. (2019). Eutrophication will increase methane emissions from lakes and impoundments during the 21st century. *Nature Communications*, 10. https://doi.org/10.1038/s41467-019-09100-5

Canelhas, M. R., Denfeld, B. A., Weyhenmeyer, G. A., Bastviken, D., & Bertilsson, S. (2016). Methane oxidation at the water-ice interface of an ice-covered lake. *Limnology & Oceanography*, 61, S78–S90. https://doi.org/10.1002/lno.10288

Chauhan, A., Pathak, A., & Ogram, A. (2012). Composition of methane-oxidizing bacterial communities as a function of nutrient loading in the Florida everglades. *Microbial Ecology*, 64(3), 750–759. https://doi.org/10.1007/s00248-012-0058-2

Crevecoeur, S., Ruiz-Gonzalez, C., Prairie, Y. T., & del Giorgio, P. A. (2019). Large-scale biogeography and environmental regulation of methanotrophic bacteria across boreal inland waters. *Molecular Ecology*, 28(18), 4181–4196. https://doi.org/10.1111/mec.15223

Davidson, T. A., Audet, J., Jeppesen, E., Landkildehus, F., Lauridsen, T. L., Sondergaard, M., & Syvaranta, J. (2018). Synergy between nutrients and warming enhances methane ebullition from experimental lakes. *Nature Climate Change*, 8(2), 156, 160. https://doi.org/10.1038/s41558-017-0063-z

Dean, J. F., Middelburg, J. J., Rockmann, T., Aerts, R., Blauw, L. G., Egger, M., et al. (2018). Methane feedbacks to the global climate system in a warmer world. *Reviews of Geophysics*, 56(1), 207–250. https://doi.org/10.1002/2017rg000559

DeBruyn, J. M., Leigh-Bell, J. A., McKay, R. M. L., Bourbonniere, R. A., & Wilhelm, S. W. (2004). Microbial distributions and the impact of phosphorus on bacterial activity in Lake Erie. *Journal of Great Lakes Research*, *30*(1), 166–183. https://doi.org/10.1016/ s0380-1330(04)70338-x

DelSontro, T., Beaulieu, J. J., & Downing, J. A. (2018). Greenhouse gas emissions from lakes and impoundments: Upscaling in the face of global change. *Limnology and Oceanography Letters*, 3(3), 64–75. https://doi.org/10.1002/lol2.10073

- Denfeld, B. A., Baulch, H. M., del Giorgio, P. A., Hampton, S. E., & Karlsson, J. (2018). A synthesis of carbon dioxide and methane dynamics during the ice-covered period of northern lakes. *Limnology and Oceanography Letters*, 3(3), 117–131. https://doi.org/10.1002/ lol2.10079
- Denfeld, B. A., Ricao Canelhas, M., Weyhenmeyer, G. A., Bertilsson, S., Eiler, A., & Bastviken, D. (2016). Constraints on methane oxidation in ice-covered boreal lakes. Journal of Geophysical Research-Biogeosciences, 121(7), 1924–1933. https://doi.org/10.1002/2016jg003382

Downing, J. A., Prairie, Y. T., Cole, J. J., Duarte, C. M., Tranvik, L. J., Striegl, R. G., et al. (2006). The global abundance and size distribution of lakes, ponds, and impoundments. *Limnology & Oceanography*, 51(5), 2388–2397. https://doi.org/10.4319/lo.2006.51.5.2388

- Duc, N., Crill, P., & Bastviken, D. (2010). Implications of temperature and sediment characteristics on methane formation and oxidation in lake sediments. *Biogeochemistry*, 100(1–3), 185–196. https://doi.org/10.1007/s10533-010-9415-8
- Dumestre, J. F., Guezennec, J., Galy-Lacaux, C., Delmas, R., Richard, S., & Labroue, L. (1999). Influence of light intensity on methanotrophic bacterial activity in Petit Saut Reservoir, French Guiana. *Applied and Environmental Microbiology*, 65(2), 534–539. https://doi. org/10.1128/aem.65.2.534-539.1999
- Garcia, S. L., Szekely, A. J., Bergvall, C., Schattenhofer, M., & Peura, S. (2019). Decreased snow cover stimulates under-ice primary producers but impairs methanotrophic capacity. *mSphere*, 4(1). https://doi.org/10.1128/mSphere.00626-18
- Gray, N. D., McCann, C. M., Christgen, B., Ahammad, S. Z., Roberts, J. A., & Graham, D. W. (2014). Soil geochemistry confines microbial abundances across an arctic landscape; implications for net carbon exchange with the atmosphere. *Biogeochemistry*, 120(1–3), 307–317. https://doi.org/10.1007/s10533-014-9997-7
- Greene, S., Anthony, K. M. W., Archer, D., Sepulveda-Jauregui, A., & Martinez-Cruz, K. (2014). Modeling the impediment of methane ebullition bubbles by seasonal lake ice. *Biogeosciences*, *11*(23), 6791–6811. https://doi.org/10.5194/bg-11-6791-2014
- Herlemann, D. P. R., Labrenz, M., Jurgens, K., Bertilsson, S., Waniek, J. J., & Andersson, A. F. (2011). Transitions in bacterial communities along the 2000 km salinity gradient of the Baltic Sea. *The ISME Journal*, 5(10), 1571–1579. https://doi.org/10.1038/ismej.2011.41

Ho, A., de Roy, K., Thas, O., De Neve, J., Hoefman, S., Vandamme, P., et al. (2014). The more, the merrier: Heterotroph richness stimulates methanotrophic activity. *The ISME Journal*, 8(9), 1945–1948. https://doi.org/10.1038/ismej.2014.74

- Iguchi, H., Yurimoto, H., & Sakai, Y. (2011). Stimulation of methanotrophic growth in cocultures by Cobalamin excreted by Rhizobia. *Applied and Environmental Microbiology*, 77(24), 8509–8515. https://doi.org/10.1128/aem.05834-11
- Kalyuzhnaya, M. G., Khmelenina, V. N., Kotelnikova, S., Holmquist, L., Pedersen, K., & Trotsenko, Y. A. (1999). Methylomonas scandinavica sp nov., a new methanotrophic psychrotrophic bacterium isolated from deep igneous rock ground water of Sweden. Systematic & Applied Microbiology, 22(4), 565–572. https://doi.org/10.1016/s0723-2020(99)80010-1
- Kankaala, P., Huotari, J., Peltomaa, E., Saloranta, T., & Ojala, A. (2006). Methanotrophic activity in relation to methane efflux and total heterotrophic bacterial production in a stratified, humic, boreal lake. *Limnology & Oceanography*, *51*(2), 1195–1204. https://doi. org/10.4319/lo.2006.51.2.1195
- Khmelenina, V. N., Makutina, V. A., Kalyuzhnaya, M. G., Rivkina, E. M., Gilichinsky, D. A., & Trotsenko, Y. (2002). Discovery of viable methanotrophic bacteria in permafrost sediments of northeast Siberia. Doklady Biological Sciences: Proceedings of the Academy of Sciences of the USSR, Biological sciences sections, 384, 235–237. https://doi.org/10.1023/a:1016025709833
- Kiersztyn, B., Chrost, R., Kalinski, T., Siuda, W., Bukowska, A., Kowalczyk, G., & Grabowska, K. (2019). Structural and functional microbial diversity along a eutrophication gradient of interconnected lakes undergoing anthropopressure. *Scientific Reports*, 9. http://10.1038/s41598-019-47577-8

#### Acknowledgments

This study was supported by the European Research Council (ERC) grant METLAKE (grant no. 725546), the Knut and Alice Wallenberg Foundation (dnr: 2016.0083), FORMAS (grant 2016-00874), the Swedish Research Council (grants 2016-04829 and 2017-04422), and the Science for Life Laboratory. The authors would like to acknowledge support from Science for Life Laboratory, the National Genomics Infrastructure, NGI, and Uppmax for providing assistance in massive parallel sequencing and computational infrastructure. We want to thank Ingrid Sundgren and Gustav Pajala for lab assistance at the Department of Thematic Studies, Linkoping University, and the anonymous reviewers and the Associate Editor for their constructive comments.



- Kozich, J. J., Westcott, S. L., Baxter, N. T., Highlander, S. K., & Schloss, P. D. (2013). Development of a dual-index sequencing strategy and curation pipeline for analyzing amplicon sequence data on the MiSeq Illumina sequencing platform. *Applied and Environmental Microbiology*, 79(17), 5112–5120. https://doi.org/10.1128/aem.01043-13
- Malashenko, Y. R., Pirog, T. P., Romanovskaya, V. A., Sokolov, I. G., & Grinberg, T. A. (2001). Search for methanotrophic producers of exopolysaccharides. *Applied Biochemistry and Microbiology*, *37*(6), 599–602. https://doi.org/10.1023/a:1012307202011
- Martinez-Cruz, K., Sepulveda-Jauregui, A., Anthony, K. W., & Thalasso, F. (2015). Geographic and seasonal variation of dissolved methane and aerobic methane oxidation in Alaskan lakes. *Biogeosciences*, 12(15), 4595–4606. https://doi.org/10.5194/bg-12-4595-2015
- Matthews, E., Johnson, M. S., Genovese, V., Du, J., & Bastviken, D. (2020). Methane emission from high latitude lakes: Methane-centric lake classification and satellite-driven annual cycle of emissions. *Scientific Reports*, 10(1). https://doi.org/10.1038/s41598-020-68246-1
- McMurdie, P. J., & Holmes, S. (2013). Phyloseq: An R package for reproducible interactive analysis and graphics of microbiome census data. *PloS One*, 8(4), e61217. https://doi.org/10.1371/journal.pone.0061217
  Muraca, L. & Sunimeta, A. (2005). Inhibitory afford of light on motherne ovidation in the polarie under column of a mesotrophic lake (Lelean existence). Inhibitory afford of light on motherne ovidation in the polarie under column of a mesotrophic lake (Lelean existence).
- Murase, J., & Sugimoto, A. (2005). Inhibitory effect of light on methane oxidation in the pelagic water column of a mesotrophic lake (Lake Biwa, Japan). *Limnology & Oceanography*, *50*(4), 1339–1343. https://doi.org/10.4319/lo.2005.50.4.1339
- R Core Team. (2019). R: A language and environment for statistical computing. R Foundation for Statistical Computing. Retrieved from http://www.R-project.org/
- Reis, P. C. J., Thottathil, S. D., Ruiz-González, C., & Prairie, Y. T. (2020). Niche separation within aerobic methanotrophic bacteria across lakes and its link to methane oxidation rates. *Environmental Microbiology*, 22(2), 738–751. https://doi.org/10.1111/1462-2920.14877
- Rissanen, A. J., Saarenheimo, J., Tiirola, M., Peura, S., Aalto, S. L., Karvinen, A., & Nykanen, H. (2018). Gammaproteobacterial methanotrophs dominate methanotrophy in aerobic and anaerobic layers of boreal lake waters. *Aquatic Microbial Ecology*, 81(3), 257–276. https://doi.org/10.3354/ame01874
- Rognes, T., Flouri, T., Nichols, B., Quince, C., & Mahe, F. (2016). VSEARCH: A versatile open source tool for metagenomics. *Peerj*, 4, e2584. https://doi.org/10.7717/peerj.2584
- Samad, M. S., & Bertilsson, S. (2017). Seasonal variations in abundance and diversity of bacterial Mathanotrophs in five temperate lakes. Frontiers in Microbiology, 8. https://doi.org/10.3389/fmicb.2017.00142
- Segers, R. (1998). Methane production and methane consumption: A review of processes underlying wetland methane fluxes. *Biogeochemistry*, 41(1), 23–51. https://doi.org/10.1023/a:1005929032764
- Sepulveda-Jauregui, A., Anthony, K. M. W., Martinez-Cruz, K., Greene, S., & Thalasso, F. (2015). Methane and carbon dioxide emissions from 40 lakes along a north-south latitudinal transect in Alaska. *Biogeosciences*, 12(11), 3197–3223. https://doi.org/10.5194/bg-12-3197-2015
- Sepulveda-Jauregui, A., Hoyos-Santillan, J., Martinez-Cruz, K., Anthony, K. M. W., Casper, P., Belmonte-Izquierdo, Y., & Thalasso, F. (2018). Eutrophication exacerbates the impact of climate warming on lake methane emission. *The Science of the Total Environment*, 636, 411–419. https://doi.org/10.1016/j.scitotenv.2018.04.283
- Sharma, S., Blagrave, K., Magnuson, J. J., O'Reilly, C. M., Oliver, S., Batt, R. D., et al. (2019). Widespread loss of lake ice around the Northern Hemisphere in a warming world. *Nature Climate Change*, 9, 227–231. https://doi.org/10.1038/s41558-018-0393-5
- Sharma, S., Meyer, M. F., Culpepper, J., Yang, X., Hampton, S., Berger, S. A., et al. (2020). Integrating perspectives to understand lake ice dynamics in a changing world. *Journal of Geophysical Research-Biogeosciences*, 125(8). https://doi.org/10.1029/2020jg005799
- Shelley, F., Ings, N., Hildrew, A. G., Trimmer, M., & Grey, J. (2017). Bringing methanotrophy in rivers out of the shadows. Limnology & Oceanography, 62(6), 2345–2359. https://doi.org/10.1002/lno.10569
- Sinclair, L., Osman, O. A., Bertilsson, S., & Eiler, A. (2015). Microbial community composition and diversity via 16S rRNA gene amplicons: Evaluating the Illumina Platform. *PloS One*, *10*(2), e0116955. https://doi.org/10.1371/journal.pone.0116955
- SMHI. (2006). Is på sjöar och älvar. Retrieved from Norrköping: https://www.smhi.se/publikationer/is-pa-sjoar-och-alvar-1.6375
- Thottathil, S. D., Reis, P. C. J., del Giorgio, P. A., & Prairie, Y. T. (2018). The extent and regulation of summer methane oxidation in Northern Lakes. *Journal of Geophysical Research-Biogeosciences*, *123*(10), 3216–3230. https://doi.org/10.1029/2018jg004464
- Thottathil, S. D., Reis, P. C. J., & Prairie, Y. T. (2019). Methane oxidation kinetics in northern freshwater lakes. *Biogeochemistry*, 143, 105–116. https://doi.org/10.1007/s10533-019-00552-x
- Trotsenko, Y. A., & Khmelenina, V. N. (2005). Aerobic methanotrophic bacteria of cold ecosystems. FEMS Microbiology Ecology, 53(1), 15–26. https://doi.org/10.1016/j.femsec.2005.02.010
- van Grinsven, S., Damste, J. S. S., Harrison, J., & Villanueva, L. (2020). Impact of electron acceptor availability on methane-influenced microorganisms in an enrichment culture obtained from a Stratified Lake. Frontiers in Microbiology, 11. https://doi.org/10.3389/ fmicb.2020.00715
- van Grinsven, S., Oswald, K., Wehrli, B., Jegge, C., Zopfi, J., Lehmann, M. F., & Schubert, C. J. (2021). Methane oxidation in the waters of a humic-rich boreal lake stimulated by photosynthesis, nitrite, Fe(III) and humics. *Biogeosciences*, *18*(10), 3087–3101. https://doi.org/10.5194/bg-18-3087-2021
- Vecherskaya, M. S., Galchenko, V. F., Sokolova, E. N., & Samarkin, V. A. (1993). Activity and species composition of aerobic methanotrophic communities in tundra soils. *Current Microbiology*, 27(3), 181–184. https://doi.org/10.1007/bf01576018
- Veraart, A. J., Steenbergh, A. K., Ho, A., Kim, S. Y., & Bodelier, P. L. E. (2015). Beyond nitrogen: The importance of phosphorus for CH<sub>4</sub> oxidation in soils and sediments. *Geoderma*, 259, 337–346. https://doi.org/10.1016/j.geoderma.2015.03.025
- Vrede, K., Vrede, T., Isaksson, A., & Karlsson, A. (1999). Effects of nutrients (phosphorous, nitrogen, and carbon) and zooplankton on bacterioplankton and phytoplankton-A seasonal study. *Limnology & Oceanography*, 44(7), 1616–1624. https://doi.org/10.4319/ lo.1999.44.7.1616
- Walter, K. M., Chanton, J. P., Chapin, F. S., Schuur, E. A. G., & Zimov, S. A. (2008). Methane production and bubble emissions from arctic lakes: Isotopic implications for source pathways and ages. *Journal of Geophysical Research-Biogeosciences*, 113. https://doi. org/10.1029/2007jg000569
- Westheimer, F. H. (1987). Why nature chose phosphates. Science, 235(4793), 1173-1178. https://doi.org/10.1126/science.2434996
- Wik, M., Johnson, J. E., Crill, P. M., DeStasio, J. P., Erickson, L., Halloran, M. J., et al. (2018). Sediment characteristics and methane ebullition in Three Subarctic Lakes. *Journal of Geophysical Research-Biogeosciences*, *123*(8), 2399–2411. https://doi.org/10.1029/2017jg004298
  Wik, M., Varner, R. K., Anthony, K. W., MacIntyre, S., & Bastviken, D. (2016). Climate-sensitive northern lakes and ponds are critical components of methane release. *Nature Geoscience*, *9*(2), 99, 105. https://doi.org/10.1038/ngeo2578
- Xing, X. H., Wu, H., Luo, M. F., & Wang, B. P. (2006). Effects of organic chemicals on growth of Methylosinus trichosporium OB3b. Biochemical Engineering Journal, 31(2), 113–117. https://doi.org/10.1016/j.bej.2006.06.001
- Zhang, T., Zhu, W., Mo, J., Liu, L., & Dong, S. (2011). Increased phosphorus availability mitigates the inhibition of nitrogen deposition on CH<sub>4</sub> uptake in an old-growth tropical forest, southern China. *Biogeosciences*, 8(9), 2805–2813. https://doi.org/10.5194/bg-8-2805-2011



Zheng, Y., Zhang, L.-M., & He, J.-Z. (2013). Immediate effects of nitrogen, phosphorus, and potassium amendments on the methanotrophic activity and abundance in a Chinese paddy soil under short-term incubation experiment. *Journal of Soils and Sediments*, 13(1), 189–196. https://doi.org/10.1007/s11368-012-0601-2

# **Reference From the Supporting Information**

Carlson, R. E. (1977). A trophic state index for lakes 1. Limnology & Oceanography, 22(2), 361-369. https://doi.org/10.4319/lo.1977.22.2.0361